Author's response to reviews

Title: Exploring comorbid use of marijuana, tobacco, and alcohol among 14 to 15-year-olds: Findings from a national survey on adolescent substance use

Authors:

Joanna White (j.white@hpa.org.nz)
Darren Walton (d.walton@hpa.org.nz)
Natalie Walker (n.walker@nihi.auckland.ac.nz)

Version: 5
Date: 11 July 2014

Author's response to reviews:

Cover letter is attached.